Author's response to reviews

Title: Epidemiological investigation of two parallel gastroenteritis outbreaks in school settings

Authors:

Kassiani Mellou (kmellou@gmail.org)
T Sideroglou (sideroglou@gmail.org)
M Potamiti-Komi (potamiti@gmail.com)
Petros Kokkinos (pkokkinos@upatras.gr)
Panos Ziros (zirster@gmail.com)
T Georgakopoulou (georgakopoulou@gmail.com)
Apostolos Vantarakis (avanta@upatras.gr)

Version: 3 Date: 5 March 2013

Author's response to reviews: see over
Dear Sir,

I send you attached our revised article entitled “Epidemiological investigation of two parallel gastroenteritis outbreaks in school settings” to be considered for publication in the BMC Public Health. It is one of the very few outbreaks occurred in community school settings in Greece and it is the first to be considered for publication. The article was corrected according to reviewers’ suggestions. Namely, the ethical statement was added as well as well the aim was added to the background.

I am looking forward to your reply.

Yours Sincerely

A. Vantarakis
Asst. Professor of Hygiene